Author's response to reviews

Title: Process evaluation of a school based physical activity related injury prevention programme using the RE-AIM framework

Authors:

Dorine CM Collard (dorine.collard@vumc.nl)
Mai JM Chinapaw (m.chinapaw@vumc.nl)
Evert ALM Verhagen (e.verhagen@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)

Version: 4 Date: 22 November 2010

Author's response to reviews:

Dear editor,

we made all changes that were required.
we added a conclusion section and renamed the introduction section into Background
we also removed all highlighting and tracking from the manuscript.

with kind regards
Dorine Collard